Author's response to reviews

Title: Hamartomas, teratomas and teratocarcinosarcomas of the head and neck: Report of 3 new cases with clinico-pathologic correlation, cytogenetic analysis, and review of the literature

Authors:

Semir Vranic (semir.vranic@gmail.com)
Samuel K Caughron (sam@caughron.com)
Slavisa Djuricic (djurisla@sezampro.yu)
Nurija Bilalovic (nurija@epn.ba)
Sadiq Zaman (sadizaman@creighton.edu)
Ismet Suljevic (ismetsul@bih.net.ba)
William M Lydiatt (wmlydiat@unmc.edu)
Jane Emanuel (emanuel@boystown.org)
Zoran Gatalica (zgatalica@pathology.creighton.edu)

Version: 4 Date: 13 November 2008

Author's response to reviews: see over
November 12, 2008

To:
Dr Mark Todd
Assistant Editor
BMC-series journals

RE: Vranic et al. _Hamartomas, teratomas and teratocarcinosarcomas of the head and neck: Report of three cases and review of the literature._

Dear Dr Todd,

We are re-submitting the final version of the manuscript entitled “Hamartomas, teratomas and teratocarcinosarcomas of the head and neck: Report of three cases and review of the literature” as requested in your email of November 11, 2008.

Comments to your requests:

1. The manuscript formatting is now according to the proposals of the Journal.
2. Abstract is re-written in line with the journal standards.
3. “Running title” is removed.
4. The name of the board is now stated at the end of the methods section.
5. Informed consent was not required by the Board.
6. “Competing interest” is also added as well as “authors’ contributions”.
7. In addition all references were corrected and cited accordingly to the BMC proposals. Figures are uploaded again in the same order as they appear in the paper.

We hope that these corrections will be satisfactory and the manuscript now is ready for publication.

Yours sincerely,

Semir Vranic, MD

Zoran Gatalica, MD, DSc